December 10, 2021

Dear editor,

Please find attached files of revised manuscript in word format

**Title:** Timing theory continuous nursing, resistance training: Rehabilitation and mental health of caregivers and stroke patients with traumatic fractures

**Author:** Yali Shen¹, Zongqun Zhang², Lijuan Zhu³, Jinghua Liu⁴

**Name of Journal:** World Journal of Clinical Cases

**Manuscript NO:** 70339

First of all, thank you for your careful guidance of this article. Revision has been made according to the suggestions of the reviewer:

**Reviewer:** 06129407

In the manuscript “Timing theory continuous nursing, resistance training: Rehabilitation and mental health of caregivers and stroke patients with traumatic fractures”, the authors tried to determine the effect of timing theory continuous care, with resistance training, on rehabilitation and mental health of caregivers and stroke patients with traumatic fractures. Currently, a large part of caring for patients is undertaken by caregivers and caregivers are under considerable pressure, and the burden of care is related to their health status. Adverse emotions, such as depression, seriously affect the recovery of patients, and the existence and influence of the burden of care is long-lasting. This study analyzed the effect of timing theory continuous nursing combined with resistance training on the rehabilitation of stroke patients with traumatic fractures and the mental health of caregivers, to provide guidance and a basis for clinical practice. The topic of this work is interesting. Their conclusion might improve stroke patients’ quality of life and mental health of both patients and their caregivers. The manuscript is well written and well organized, and authors presented also the limitations of the study. I recommend that the manuscript
Dear Sir, I read with interest the manuscript entitled "Timing theory continuous nursing, resistance training; Rehabilitation and mental health of caregivers and stroke patients with traumatic fractures". The manuscript is well designed and written. The introduction gives a good overview about the topic and the procedures are precisely described. The results were well discussed. However, some issues must be addressed: 1. The background introduction of this research is too brief, please add some details about traumatic fracture after stroke.; 2. The discussion section is very general and should discuss the results of this present study more precisely; 3. Finally, the reviewer also suggested that authors can add a Table to describe the baseline data more clearly.

Thank you for your advice.

After receiving the comments:
(1) We have added the details of traumatic fractures after stroke in the background part;
(2) We refined the discussion of the results of this research in the discussion section;
(3) We have added a table describing the baseline data;
(4) We have corrected some minor language errors.

Thank you again for publishing our manuscript in the World Journal of Clinical Cases.

Sincerely Yours,

Jinghua Liu